Author's response to reviews

Title: Estrogen Receptor-Beta Gene Polymorphism in women with Breast Cancer at the Imam Khomeini Hospital Complex, Iran

Authors:

Sakineh Abbasi (sakineh4612004@yahoo.com)
Cyrus Azimi (cazimi@yahoo.com)

Version: 2 Date: 3 May 2010

Author's response to reviews: see over
Covering Letter

Dear Jenny Leigh,

I would like to inform you that all necessary corrections have made on the manuscript as follow:
1-An English editor reviewed the manuscript for language corrections.
2-Mistakes in the codon number corrected.
3-Information in the text corrected, since it was the first study in ER-β polymorphism in Iran, therefore there is no information available.
4-These two exons were selected randomly.
5-The study of the other exons in the same gene is the part of our next project.
6-The number of tables were corrected and the Figure was replaced by three more tables (3, 4 and 5)
7-Answer to the special question of Dr. Ivana Beatrice da Cruz: 
   Odd ratio (OR) has been calculated from comparing the normal group with each of the heterozygous and homozygous groups. Considering Table 6, row 1 in relation to breast cancer, for the absence of heterozygous and homozygous there is no possibility of calculation of OR, therefore not mentioned.
8-The parts of Discussion and Conclusion completely revise according to the suggestions.
9-Introduction paragraphs, 3, 4 and 5 were revised according to suggestions.
10- The authors’ contributions added to the text in directed format.

Best Regards
Dr. Sakineh Abbasi